Author's response to reviews

Title: Mobile phones improve antenatal care attendance and quality of care in Zanzibar: A cluster randomized controlled trial

Authors:

Stine Lund (stine_lund@dadlnet.dk)
Birgitte B Nielsen (birgitte.bruun.nielsen@dadlnet.dk)
Maryam Hemed (hemed32@yahoo.com)
Ida M Boas (idaboas@yahoo.com)
Azzah Said (azzahnofly@yahoo.co.uk)
Khadija Said (khadijasd@yahoo.com)
Mkoko H Makungu (makungu@yahoo.com)
Vibeke Rasch (vrasch@health.sdu.dk)

Version: 3
Date: 2 January 2014

Author's response to reviews: see over
It is with pleasure that we submit a revised version of our article “Mobile phones improve antenatal care attendance in Zanzibar: A cluster randomized controlled trial”.

We would like to thank editor and reviewers for their comments and input to further improve the article. Please find below a point-to-point description of the changes made.

Yours sincerely,

Stine Lund, MD, PhD Fellow

Editorial Request

Please include a title page at the front of your manuscript file. It should contain, at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.

*The title page has been expanded with names and email addresses of all authors.*

Reviewer 1

1. My comments have been addressed and the manuscript looks much better

   Thank you.

2. Corrected figures in Table 2 on antepartum referral require also correction in Table 4.

   Thank you; the figures in Table 4 have been corrected.